Author’s response to reviews

Title: Knowledge and attitude towards pregnancy-related issues of Zika virus infection among general practitioners in Indonesia

Authors:

Harapan Harapan (harapan@unsyiah.ac.id)
Yogambigai Rajamoorthy (yogambigai@utar.edu.my)
Prattama Utomo (prattamasantoso@gmail.com)
Samsul Anwar (s4m_seupeng@yahoo.co.id)
Abdul Setiawan (drmalique@gmail.com)
Alma Alleta (alma@unsyiah.ac.id)
Alfredo Bambang (alfredoibc2012@gmail.com)
Muhammad Ramadana (rizki.mpawill@gmail.com)
Ikram Ikram (ikramkeren@yahoo.co.id)
Nur Wahyuniati (nur.wahyuniati@unsyiah.ac.id)
Reza Maulana (rezamaulana@unsyiah.ac.id)
Ichsan Ichsan (ichsanmd_aceh@unsyiah.ac.id)
Rosaria Indah (rosariaindah1974@gmail.com)
Abram Wagner (awag@umich.edu)
Ulrich Kuch (kuch@med.uni-frankfurt.de)
Alfonso Rodríguez-Morales (ajrodriguezmmmd@gmail.com)
David Groneberg (arbsozmed@uni-frankfurt.de)
Mohd Andalas (andalas@unsyiah.ac.id)
Ruth Müller (ruth.mueller@med.uni-frankfurt.de)
Mudatsir Mudatsir (mudatsir@unsyiah.ac.id)
Author’s response to reviews:

Dear Editor,

Thank you for giving us opportunity to revise our manuscript. We have revised it based on Editor comments. Here are our response to each

1. It has come to our attention that within your manuscript there is significant text overlap with other publications,

   Background:


   While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please re-phrase this section in your own words to avoid direct overlap (please note, we cannot accept direct textual overlap with any previously published works even if the authors of the work are yourselves). Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated.

   RESPONSE: Thank you for your diligence in checking and we apologize for the text overlap. We have made edits throughout the Introduction.

2. In the “Funding” section of your declarations, please clarify the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

   RESPONSE: We have added statement the role of funders: "The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript". (Please see Funding).
3. Please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this in your "Ethics and Consent to Participate" section. If verbal, please state the reason and whether the ethics committee approved this procedure.

RESPONSE: We have added statement related to this concern: "This was an electronic survey. Potential participants were first shown an informed consent document and were instructed to click "Yes" if they agreed to the procedures. As per the protocol approved by the ethical review committee, individuals' electronic signatures were not collected in order to limit the amount of personally identifiable information held by the researchers. No direct incentive was given to participate in this survey." (Please see Ethics approval and consent to participate)

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

RESPONSE: We now providing clean manuscript only with Tables and Additional files uploaded in separate files.

Thank you

Best regards,

Harapan